Author’s response to reviews

Title: Defective angles of localized retinal nerve fiber layer reflect the severity of visual field defect- a cross-sectional analysis

Authors:

Alexander Chen (a1050276@hotmail.com)
Ing-Chou Lai (lai1@cgmh.org.tw)
Wan-Hua Cho (a725157@cgmh.org.tw)
Hung-Yin Lai (duckmelissa@hotmail.com)
Pei-Wen Lin (lpw324@cgmh.org.tw)
Pei-Chang Wu (wpc@cgmh.org.tw)
Ming-Tse Kuo (mingtse@cgmh.org.tw)

Version: 3 Date: 14 Mar 2020

Author’s response to reviews:

Response to comments of the editors and reviewers

Defective angles of localized retinal nerve fiber layer reflect the severity of visual field defect- a cross-sectional analysis” (BMC Ophthalmology, BOPH-D-19-00297)

We highly appreciate the valuable comments and suggestions of the reviewers. Please see below for the point-by-point responses:

Editor comments:

Q1: Availability of data and materials

As the file linked in your data availability statement contains raw data, and therefore does not contain critical information for understanding your study, please remove it from your manuscript to ensure that participant anonymity is not compromised. At your discretion, you may instead state the raw data available upon reasonable request from the corresponding authors in the “Availability of data and materials” section of the declarations.
A1: We have removed the link and have revised the statement as follows:

“The datasets analysed during the current study are available upon reasonable request.”

Page 11 Line 227

Q2: Clean copy

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

A2: The manuscript is now single, final, and clean without any tracked changes or highlights.

Q3: Manuscript body

Please rename "Introduction" to "Background" in the manuscript body.

A3: Introduction has been renamed to background in the manuscript body.

Page 3 Line 42

Q4: Abbreviations

A4: "Abbreviations and Acronyms" has been renamed to "Abbreviations" and moved after conclusion section of the main manuscript.

Page 10 Line 211

Q5: Authors contributions

A5: The statement “all authors have read and approved the manuscript” has been included in the Authors' contributions section.

Page 11 Line 239